
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Tue Apr 15 16:48:09 GMT 2003
Year: 03  Doy: 105
Observer: koon
WEATHER COMMENT: Tue Apr 15 16:48:10 GMT 2003
Thicl cirrostratus overcast, wind=5 mph from the SE, temp=42 F.
COMMENT: Tue Apr 15 16:50:00 GMT 2003
All workstations took a lot longer than usual to allow me to login this morning
and Kii shows the message: " NFS server Kaiee not responding.  Bunny left a
phone message that she is having problems rebooting Kaiee too, Kaiee shows
a login prompt.  There have been data transfer problems with Kaiee since last
Thursday too - according to what I have read this morning.
COMMENT: Tue Apr 15 17:00:41 GMT 2003
Will wait for clouds to thin before starting instrumants.
COMMENT: Tue Apr 15 17:27:37 GMT 2003
Kaiee is back up and running normally, I had to do a Stop-A and then "boot" to
get it rebooted, it took a long time to reboot but things seem normal after
the reboot.
A NEW TAPE HAS BEEN PUT INTO KAIEE DLT DRIVE, Tue Apr 15 17:59:10 GMT 2003
Tue Apr 15 20:13:40 GMT 2003    CHIP     Start Patrol
Tue Apr 15 20:13:53 GMT 2003    PICS     Start Patrol
WEATHER COMMENT: Tue Apr 15 20:13:38 GMT 2003
Clouds have thinned enough for the spar to track now and then, starting
some instruments.
PSPT COMMENT: Tue Apr 15 21:35:27 GMT 2003
Observing.
Tue Apr 15 22:52:55 GMT 2003    CHIP     End Patrol
WEATHER COMMENT: Tue Apr 15 22:53:20 GMT 2003
Clouds are very thick again, spar not guiding, pausing instruments.
Tue Apr 15 22:53:50 GMT 2003    PICS     End Patrol
Wed Apr 16 00:25:13 GMT 2003    CHIP     Start Patrol
Wed Apr 16 00:25:26 GMT 2003    PICS     Start Patrol
WEATHER COMMENT: Wed Apr 16 00:25:12 GMT 2003
Spar is barely tracking again, restarting some instruments.  Still have thick
cirrostratus overcast.
Wed Apr 16 02:06:42 GMT 2003    CHIP     End Patrol
Wed Apr 16 02:06:44 GMT 2003    PICS     End Patrol
Wed Apr 16 02:15:46 GMT 2003 
        MkIV    
 


